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В настоящей работе приводятся результаты исследования зависимости скорости конвекции жидкости в микропробирке от угла
наклона при точечном нагреве. При проведении данного исследования используется ранее подготовленная конечно-объемная
сетка конусообразной микропробирки для использования в рамках программного пакета OpenFOAM. Проведены подробный ана-
лиз и описание математической модели, а также серия вычислительных экспериментов с использованием встроенного решателя
buoyantBoussinesqPimpleFoam.

Важным этапом данной работы является пост-обработка результатов, полученных в ходе проведения серии вычислительных
экспериментов. В работе приведено подробное описание особенностей проведенной обработки данных для прямого количественного
сравнения результатов, полученных в отдельных экспериментах серии.

В проведенных вычислительных экспериментах были получены количественные данные для выявления зависимости скорости
конвекции жидкости в микропробирке от угла наклона при точечном нагреве. На основе полученных результатов выявлено суще-
ственное влияние угла наклона на скорость конвекции, что в дальнейшем может послужить решением для проблем оптимизации
процессов перемешивания в микромасштабных системах.

Ключевые слова: полимеразная цепная реакция, OpenFOAM, температурная конвекция, приближение Буссинеска, вычисли-
тельное моделирование

Modeling of temperature convection in microtubes under point heating:
dependence of convection velocity on tilt angle
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This paper presents the results of a study of the dependence of the convection velocity of a liquid in a microtube on the angle of inclination
during point heating. This study utilizes a previously prepared finite-volume mesh of a cone-shaped microtube for use within the OpenFOAM
software package. A detailed analysis and description of the mathematical model and a series of computational experiments using the
built-in buoyantBoussinesqPimpleFoam solver are performed.

An important step in this paper is the post-processing of the results obtained from a series of computational experiments. The
paper provides a detailed description of the features of the data processing carried out for direct quantitative comparison of the results
obtained in the individual experiments of the series.

In the conducted computational experiments, quantitative data were obtained to reveal the dependence of the liquid convection
velocity in a microtube on the angle of inclination during spot heating. From the obtained results, a significant influence of the inclination
angle on the convection velocity was revealed, which can serve as a solution for the problems of optimization of mixing processes
in microscale systems in the future.
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1. Введение

Полимеразная цепная реакция (ПЦР) — это самый
распространенный метод увеличения объема нуклеи-
новых кислот. ПЦР широко используется в молекуляр-
ной биологии для увеличения количества ДНК в образ-

цах и проведения генетических анализов (секвенирова-
ния генома). Этап ускорения играет ключевую роль в
процессе ПЦР, поскольку определяет начальные усло-
вия для последующей амплификации ДНК. Понима-
ние и оптимизация этого этапа имеют большое зна-
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чение для достижения точных и надежных результатов.
Впервые ПЦР была описана Нобелевским лауреатом
Кэри Маллисом в 1983 году [1].

В 1996 году компания Applied Biosystems предста-
вила метод количественной ПЦР с обратной транскрип-
цией (ПЦР-ОТ) в виде общедоступной технологии, вы-
пустив на рынок прибор 7700 [2]. В то время этот ме-
тод стал наиболее точным и чувствительным способом
обнаружения и количественного определения нуклеи-
новых кислот.

В связи с необходимостью оптимизации процесса
диагностики ДНК большое внимание уделяется сокра-
щению продолжительности ПЦР. Существует несколько
способов ускорения ПЦР. Наиболее широко используе-
мым методом является проведение конвективной ПЦР.
В этом случае изменение температуры реакционной
смеси достигается за счет перемещения реакционно-
го содержимого под действием тепловой конвекции.
Движение жидкости осуществляется за счет темпера-
турного градиента, возникающего из-за нагрева и (или)
охлаждения некоторых точек микропробирки, в кото-
рой осуществляется ПЦР [3].

Первые упоминания о роли конвективных сил в
микрофлюидных устройствах для амплификации нук-
леиновых кислот встречаются в патентной литературе.
В патенте США [2], описывающем кремниевое микро-
флюидное устройство для проведения реакций с нукле-
иновыми кислотами (включая ПЦР и лигазную цепную
реакцию), термоконвекция используется для пассивно-
го охлаждения реакционной смеси у стенок устройства.

В работе [4], посвященной разработке метода на-
грева реакционной смеси за счет электролитического
сопротивления, также упоминается охлаждение посред-
ством принудительных конвекционных потоков возду-
ха.

В отечественной литературе [5] также встречают-
ся упоминания о конвективных потоках в контексте
проведения ПЦР в массивном термоблоке, однако от-
мечается их незначительная интенсивность.

Первое описание «истинной» конвекционной ПЦР,
где конвективные токи выступают в качестве основ-
ной движущей силы, перемешивающей слои жидкости,
представлено в патенте [6].

В работе [7] представлена новая концепция прове-
дения ПЦР с использованием конвекционной ячейки,
работающейна основе принципа Бенара–Рэлея. Устрой-
ство выполнено в виде кубика из оргстекла с вертикаль-
ным каналом, имеющим глубину 1,5 см и объем 35 мкл.
Нагрев нижней части кубика до 97◦С и термостатирова-
ние верхней части при 61◦С обеспечивают циркуляцию
реакционной смеси, имитируя температурные циклы
ПЦР. Экспериментально подтверждена работоспособ-
ность предложенной системы: за 1,5 часа реакции был
получен специфичный ПЦР-продукт, детектируемый
методом агарозного гель-электрофореза.

Несмотря на попытки усовершенствования конвек-
ционной ПЦР-системы, предпринятые авторами рабо-
ты [8], сохранялись основные недостатки данного ме-

тода, такие как низкая скорость реакции и сложности с
заполнением реакционных сосудов. Время инкубации,
необходимое для получения количества целевого про-
дукта, сопоставимого с традиционной ПЦР, составляло
около 40 минут.

В работе [9] авторы предложили альтернативный
вариант конвекционной ячейки для ПЦР, добившись
при этом впечатляющей скорости амплификации —
около 10 минут — за счет сокращения времени од-
ного цикла (денатурация–отжиг–элонгация) до 15 се-
кунд. Однако, реакционный сосуд в виде тонкого
слоя жидкости между покровными стеклами, герме-
тизированными силиконом, делает данную систему
непригодной для широкого применения, особенно
в области ДНК-диагностики.

Анализ существующих исследований в области тер-
моконвекции демонстрирует, что формирование кон-
вективных ячеек различных типов неизменно связыва-
лось с созданием температурного градиента.

Так, в большинстве работ нагрев осуществлялся
снизу, что обеспечивало вертикальный градиент темпе-
ратур. В других исследованиях воздействие различных
температур осуществлялось на боковые стороны систе-
мы, что приводило к возникновению горизонтального
градиента температур.

Несмотря на значительный прогресс в исследова-
нии конвективных явлений, влияние ориентации тем-
пературного градиента относительно вектора гравита-
ции на морфологию и динамику конвективных ячеек
остается недостаточно изученным. Традиционно счи-
талось, что нагрев сверху не может привести к разви-
тию термоконвекции в силу стабилизирующего воздей-
ствия гравитации.

Однако развитие экспериментальных методов, в
частности, использование лазерных систем для локаль-
ного нагрева жидкости, позволило инициировать кон-
вективные течения и при инвертированном темпера-
турном градиенте. Например, фокусировка лазерно-
го излучения в центре емкости с жидкостью приводит
к возникновению восходящих конвективных потоков,
что нашло практическое применение в реализации кон-
векционной ПЦР.

Дальнейшие исследования в этом направлении
представляют значительный интерес для углубленно-
го понимания механизмов формирования и эволюции
конвективных структур в различных условиях.

За прошедшие почти два десятилетия существова-
ния конвекционной ПЦР эта реакция доказала свою
жизнеспособность и в настоящее время разработаны
различные варианты метода с одним, двумя или даже
тремя источниками нагрева и разнообразной формой
реакционных сосудов.

Исследование проблем ускорения ПЦР остается ак-
туальным и важным даже при наличии уже разработан-
ных средств и методов. Экспериментальные проверки
требуют больших трудозатрат, в связи с чем появляется
необходимость в предварительном моделировании и
проведении расчетов для дальнейших исследований.
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В настоящей работе на основе вычислительного
моделирования рассматривается предположение, что
помимо температурного градиента существенное вли-
яние на время проведения ПЦР может оказывать также
и угол наклона микропробирки, то есть изменение гео-
метрии области конвекции относительно вертикали.

2. Температурная конвекция
Существует два вида конвекции: естественная (сво-

бодная) и вынужденная. Естественная конвекция возни-
кает из-за разности плотностей жидкости, вызванной
ее контактом с поверхностью с другой температурой,
что вызывает подъемные (архимедовы) силы. Примера-
ми естественной конвекции являются теплоотдача от
стен или крыши здания в безветренный день, конвек-
ция в сосуде с жидкостью, содержащей нагревательную
спираль, или теплоотдача от солнечного коллектора в
штиль.

Вынужденная конвекция возникает, когда жид-
кость обтекает поверхность с другой температурой под
действием внешней силы. Скорость жидкости при на-
сильственной конвекции выше, чем при свободной, что
позволяет передавать больше тепла при том же темпе-
ратурном градиенте. Однако эта увеличенная переда-
ча тепла компенсируется энергией, затрачиваемой на
поддержание движения жидкости. Вне зависимости от
вида конвекции количество тепла, переносимого теп-
ловым потоком, можно вычислить с помощью закона
Ньютона–Рихмана:

∂ρ

∂t
+ ∇ · (ρu) = 0.

Представления о механизмах теплопереноса в раз-
личных средах начали формироваться задолго до воз-
никновения строгих научных теорий. В частности, ан-
глийский метеоролог Дж. Гадлея (G. Hadley) в 1735 го-
ду и российский ученый М.В. Ломоносов в 1753 году
выдвигали гипотезы о существовании таких механиз-
мов. Важно отметить, что сам феномен температур-
ной конвекции в природных системах был эмпири-
чески известен еще со времен античности, начиная
с трудов Архимеда [10].

Возможно первое практическое применение теп-
ловой конвекции состоялось в середине XVIII века бла-
годаря англичанину полковнику W. Cook, предложив-
шему для согрева помещений использовать проложен-
ные вдоль стен комнат металлические трубы, в кото-
рые подавался водяной пар из специальной емкости,
нагреваемой на кухне [11].

На рубеже XIX–XX веков были опубликованы осно-
вополагающие труды французского ученого Г. Бенара
(H. Benard), посвященные систематическому изучению
феномена термоконвекции. Эти работы положили на-
чало активным исследованиям в данной области.

3. Микрофлюидика
Первыми результатами в микрофлюидике (мик-

ромасштабные течения) следует считать обнаружение

конвективных ячеек Рэлея–Бенара. Несмотря на то,
что Рэлей [12] предсказал формирование конвективных
ячеек под действием градиента плотности, вызванного
неоднородным нагревом, теоретическое развитие этой
концепции, предпринятое Джеффрисом(H. Jeffreys) [13],
не получало экспериментального подтверждения до
1935 года [14]. Обнаружение данного феномена не было
признано научным сообществом того времени вслед-
ствие распространенности ошибочных представлений
о конвекции, основанных на авторитетном мнении Рэ-
лея.

Однако микрофлюидика как научное направление,
исследующее поведение жидкостей и газов на микро-
уровне, возникла в 1990-х годах. Современным этапом
ее развития стала нанофлюидика [15], изучающая яв-
ления в наноразмерных системах, транспорт молекул
через наноканалы, взаимодействие с наноструктурами
и другие аспекты.

Бурное развитие микрофлюидики привело к появ-
лению инновационных устройств, способных контро-
лировать потоки жидкости и газожидкостные потоки
объемом нано- и пиколитров в микроскопических ка-
налах с высокой точностью [16]. Эта технология откры-
вает новые возможности для аналитических методов
и разработки устройств с уникальными техническими
характеристиками, не имеющими аналогов в макро-
масштабе.

Впервые возможность ПЦР на микрочипах бы-
ло продемонстрировано в работах исследовательской
группы под руководством А. Нортропа [17]. В 1994 году
были опубликованы работы исследователей под руко-
водством Рэмзи, в которых представлены новые ме-
тоды разделения образцов на микрофлюидных чипах
с применением электрофореза и электрохроматогра-
фии [18].

Приборы на основе микрофлюидных чипов име-
ют значительные преимущества по сравнению с тра-
диционными аналитическими системами: экономия
реагентов и образцов, высокая чувствительность обна-
ружения компонентов образца, компактные размеры
и низкое энергопотребление.

Микроустройства для проведения ПЦР могут быть
разделены на группы в зависимости от принципа и ме-
ханизма нагревания реакционной смеси (прямое нагре-
вание камеры, нагрев смеси в потоке, конвекционное
нагревание, нагревание электромагнитным излучени-
ем) и способов перемещения реакционной смеси (ста-
ционарныеипроточные системы) [19]. К стационарным
системам с реакционными камерами, где температура
меняется в каждом цикле амплификации, относятся
однокамерные чипы [20] и мультикамерные чипы [21].

4. Математическая модель
Основной причиной тепловой конвекции является

изменение плотности среды в зависимости от темпе-
ратуры. В большинстве случаев изменения плотности
невелики, и их можно учитывать только в члене, отвеча-
ющем за подъемную силу. Для вывода уравнений тепло-
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вой конвекции используется уравнение Навье–Стокса,
дополненное уравнениями теплопроводности, закона
сохранения энергии и уравнением состояния. Этот под-
ход дает систему нелинейных уравнений в частных про-
изводных, которуюможно найти во многих источниках,
например, в [23]. Полученные уравнения называются
уравнениями тепловой конвекции в приближении Бус-
синеска [24] и при описании конвекции в несжимаемой
жидкости являются наиболее популярной моделью.

В литературе существуют разногласия в опреде-
лении термина «приближение Буссинеска» из-за того,
что уравнения конвекции были выведены Ж. Буссине-
ском только для случая уменьшения плотности жидко-
сти от температуры. Например, Моллендорф и Джен
(J.C. Mollendorf, K.H. Jahn) [25] говорят о двух аспек-
тах приближения Буссинеска: первый подразумевает
незначительные изменения плотности, а второй — ли-
нейную зависимость плотности от температуры.

В исследовании Гебхарта и соавторов (B. Gebhart et
al.) [26] за приближение Буссинеска принимается систе-
ма уравнений, описывающая линейную зависимость
плотности от температуры. В отечественной литерату-
ре приближение Буссинеска определяется как любые
приближения, в которых изменения плотности учиты-
ваются только в члене, отвечающем за подъемную силу,
который может включать в себя не только температуру,
но и, например, давление [27], ускорение свободного
падения во времени [23] и другие факторы [28].

В потоках, сопровождающихся теплопередачей,
свойства жидкости обычно зависят от температуры.
Вариации могут быть незначительными, но все же яв-
ляться причиной движения жидкости. Если изменение
плотности невелико, то плотность можно рассматри-
вать как постоянную в нестационарном и конвектив-
ном членах и как переменную только в поле внешних
сил (гравитационном члене).

В присутствии силы тяжести тела уравнения сохра-
нения массы и импульса имеют вид:

∂ρ

∂t
+ ∇ · (ρu) = 0; (1)

∂(ρu)
∂t

+ ∇ · (ρuu) = −∇p +

+ ρg + ∇ ·
(

2µe f fτττ
)

− ∇
(

2
3
µe f f (∇ · u)

)
,

τττ = 1
2

(
∇u + (∇ · u)T

)
,

(2)

где u — вектор скорости; p — давление; ρ — плот-
ность жидкости при нормальных условиях (базовая
плотность); g — вектор ускорения свободного падения;
µe f f — сумма молекулярной и турбулентной вязкостей;
τττ — тензор скорости деформации.

Если и плотность, и гравитационное ускорение по-
стоянны, гравитационная сила может быть выражена
через градиент:

ρg = ∇(ρg · r),

где r — радиус-вектор (координата).

Таким образом, градиент давления и сила тяжести
могут быть объединены в следующее уравнение:

∇p − ρg = ∇(p − ρg · r).

Теперь рассмотрим случай, когда плотность непо-
стоянна. Приближение Буссинеска справедливо, когда
изменение плотности, вызванное изменением темпе-
ратуры, невелико.

В потоках, сопровождающихся теплопередачей,
свойства жидкости обычно зависят от температуры. Ва-
риации могут быть небольшими, но все же являться
причиной движения жидкости. Если изменение плот-
ности невелико, то плотность можно рассматривать как
постоянную в нестационарном и конвективном членах
и как переменную только в гравитационном члене. Это
и называется приближением Буссинеска [24].

Здесь и далее обозначим эталонную плотность че-
рез ρ0 при эталонной температуре T0. Если мы заме-
ним ρ на ρ0 в уравнениях (1) и (2), то получим уравнение
неразрывности

∇ · u = 0

и уравнение сохранения импульсов

∂u
∂t

+ ∇ · (ρ0uu) = −∇p + ρg + ∇(2µe f fτττ). (3)

Затем разделим обе части уравнения (3) на ρ0 и по-
лучим:

∂u
∂t

+ ∇(uu) = − 1
ρ0

(∇p − ρg) + ∇ · (2µe f fτττ).

.
Здесь плотность ρ в гравитационном члене вычис-

ляется как линейная функция температуры:

ρ ≈ ρ0[1 − β(T − T0)],

β = − 1
ρ

∂ρ

∂T
≈ − 1

ρ0

ρ− ρ0

T − T0
,

где β—коэффициент температурного расширения; T0—
значение температуры для нормальных условий, при
которых плотность жидкости имеет значение ρ0; темпе-
ратура T вычисляется из уравнения теплопроводности
с учетом конвекции:

∂T
∂t

+ ∇ · (uT) = ∇ · (κ∇T), (4)

при этом диффузным членом с коэффициентом тепло-
проводности κ можно пренебречь.

С точки зрения реализации члены градиента давле-
ния и силы тяжести переставляются в следующем виде:

−∇
(

p
ρ0

)
+

(
p
ρ0

)
g =

= −∇
(

p − ρg · r
ρ0

+ ρg · r
ρ0

)
+

(
p
ρ0

)
g =

= −∇prgh − (g − r)∇
(

p
ρ0

)
,

prgh = (p − ρg · r).
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Откуда вытекает уравнение

∂u
∂t

+∇(uu)=−∇prgh − (g · r)∇
(

p
ρ0

)
+∇·(2µe f fτττ). (5)

В решателе OpenFOAM buoyantBoussinesq-
PimpleFoam решаются уравнения (4) и (5). Данные
уравнения описывают процессы, протекающие при
тепловой конвекции в микропробирках, которые были
смоделированы по результатам проведенного ранее
исследования.

Схема расчетной области с указанием точек при-
ложения нагрева и охлаждения приведена на рис. 1.
Данная схема нагрева/охлаждения задана с учетом экс-
периментов, описанных в работах [29–31]. На рисунке
также указаны основные параметры вычислительной
области: D — диаметр моделируемой части пробирки;
r — радиус закругления дна; h — высота моделируемой
части пробирки без учета дна; H — общая высота.

Граничные условия для рассматриваемой задачи
заданы в достаточно простом виде. По всем границам
принято наличие твердой стенки, что означает приме-
нение граничного условия прилипания (вектор скоро-
сти равен 0):

∇T · n = 0.

Для температуры выделены две диагонально рас-
положенные граничные области в нижней и верхней
частях пробирки, в которых температура задается как
величина с постоянным значением: область нагрева
при Thot = 400 K и область охлаждения Tcold = 300 K, на-
чальная температура принята равной 300 K. На осталь-
ных границах для температуры задан нулевой градиент
или «условие протекания»: −→v = 0.

При такой постановке задача характеризуется без-
размерными числами: число Рейнольдса — Re ≈ 10,
число Рэлея — Ra ≈ 106, число Нуссельта — Nu ≈ 1.7.

wall

wall

hot

cold

x

z

y

D = 4 мм;
r = 1 мм;
h = 7 мм;

H = 8 мм;

Рис. 1. Схема вычислительной области, представленная в сече-
нии Oxz

5. Инструменты вычислительного
моделирования
В качестве инструмента для решения поставленной

задачи выбран пакет прикладных программOpenFOAM
(Open Source Field Operation And Manipulation) [32].
OpenFOAM—это доступная система для численного мо-
делирования задач механики сплошных сред. Базовая
версия пакета OpenFOAM поставляется с множеством
программных решений, вспомогательных инструмен-
тов и библиотек. Помимо программных решений пакет
включает в себя широкий спектр утилит, позволяющих
генерировать и преобразовывать расчетные сетки, уста-
навливать значения полей и многое другое.

OpenMPI (Open Message Passing Interface) — это
стандарт для передачи сообщений и параллельных вы-
числений, который обеспечивает возможность взаимо-
действия между процессами, работающими на различ-
ных узлах кластера или на одном компьютере, и пред-
ставляющий собой набор библиотек и инструментов
для разработки параллельных приложений, которые
могут эффективно использовать вычислительные ре-
сурсы нескольких узлов [33].

Принцип работы OpenMPI включает в себя следу-
ющие аспекты:

1. Инициализация и завершение.

2. Создание коммуникаторов.

3. Обмен сообщениями.

4. Синхронизация.

5. Управление ресурсами.

Это основные принципы работы OpenMPI, кото-
рые позволяют разработчикам создавать параллельные
приложения, эффективно использующие возможности
распределенных и многопроцессорных систем.

С помощью встроенной утилиты decomposePar [33]
осуществили декомпозицию сетки и полей. В нашем
случае запуск производился на локальной машине с
многоядерным CPU 11th Gen Intel(R) Core(TM) i7-11700
@ 2.50GHz с 8 ядрами, но в конфигурации мы выстави-
ли использование только 4 ядер CPU. По итогу получили
4 подкаталога с номерами процессора. В каждом катало-
ге содержится каталог времени, содержащий описание
фрагментированных (декомпозированных) полей и ка-
талог с описанием декомпозированной сетки.

Далее проведена постобработка данных. По окон-
чанию вычислений собираем (агрегируем) данные во-
едино для последующей обработки в нормальном режи-
ме и просмотра решений в среде paraFoam. Восстанов-
ление решения осуществляется при помощи утилиты
reconstructPar [33].

Использование пространственной декомпозиции
позволяет пропорционально ускорять вычисления, про-
изводимые в рамках пакета OpenFOAM, что положи-
тельно сказывается при моделировании «тяжелых»z за-
дач динамики сплошной среды.
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6. Результаты и обсуждение
Проведена серия вычислительных экспери-

ментов с использованием встроенного решателя
buoyantBoussinesqPimpleFoam, конечно-объемной сет-
ки и граничных условий. В качестве начальных условий
в каждом расчете использовалась неподвижная жид-
кость при постоянной температуре T0 = 300 К. Расчеты
сделаны при разных направлениях вектора ускорения
свободного падения (угол наклона микропробирки) от
0◦ (ось пробирки вертикальна) до 90◦ (ось пробирки
горизонтальна). При этом поворот микропробирки
осуществляется таким образом, что зона нагрева распо-
лагается внизу закругления острого конца пробирки, а
зона охлаждения — наверху противоположного конца.

Результаты моделирования продемонстрированы
на рис. 2–4.

На рис. 2 показаны линии тока для вертикально
размещенной микропробирки. Следует отметить, что
аналогичная картина течения может быть показана и
для наклоненной пробирки. Некоторое отличие в ин-
тенсивности линий тока на рисунке не слишком раз-
личимо. Однако при некоторых положения пробирки
картина течения кардинально изменяется, как это пока-
зано на рис. 3. Такое распределение вихрей возникает
при наклоне приблизительно 11◦. На рисунке цвет ли-
ний характеризует температуру жидкости, а длина —
скорость.

После проведения расчета необходимо было обра-
ботать данные для дальнейшего анализа смоделиро-
ванного процесса, а также исследования зависимости
скорости конвекции жидкости в микропробирке от угла
наклона при точечном нагреве.

В качестве инструмента для постобработки бы-
ла использована встроенная утилита OpenFOAM —
postProcess. Данная утилита использована для вычис-
ления модуля вектора скорости в каждой ячейке сет-
ки, для получения этих данных был использован пара-
метр — mag(U). Также для дальнейшего анализа нам
потребовалось получение значения объема ячеек, для
этого был использован параметр writeCellVolumes.

После запуска следующей утилиты в качестве ре-
шения получили список/каталог по временным меткам
(от 0 до 1.1 с шагом 0.01), внутри каждого каталога на-
ходился файл с полученными данными модуля вектора
скорости в каждой ячейке сетки и объемом. Скорость
соотносится с объемом по порядку.

Для анализа интенсивности перемешивания жид-
кости, важного для задачи ускорения ПЦР, нами введе-
но понятие коэффициента перемешивания ξ. Данный
коэффициент рассчитывается из результатов модели-
рования как сумма произведений объема и значения
модуля скорости для всех конечных объемов расчетной
области. На рис. 4 представлена зависимость ξ(α), то
есть коэффициента перемешивания от угла наклона
микропробирки:

ξ(α) =
n

∑
1

(Vi · vi),

где ξ— коэффициент перемешивания; α— угол накло-
на пробирки; Vi — значение объема в i-ой ячейке; vi —
значение модуля вектора скорости в i-ой ячейке; n —
количество ячеек сетки.

Исходя из полученных результатов видно, что мак-
симальная оценка скорости тепловой конвекции дости-
гается при угле наклона α = −40◦. Кроме того, приве-
денная выше зависимость, представленные на рис. 4,
согласуется с результатами исследований, полученных
в ходе реального эксперимента [3].

Рис. 2. Визуализация линий тока для случая, когда пробирка
не наклонена

Рис. 3. Визуализация линий тока для случая, когда пробирка в
наклонном положении с углом наклона 11◦
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Рис. 4. Зависимость уровня перемешивания от угла наклона пробирки

При угле наклона α = −11◦ наблюдается наимень-
шая оценка скорости тепловой конвекции. Одной из
причин снижения скорости тепловой конвекции явля-
ется то, что области нагрева/охлаждения находятся друг
над другом, вызывая тем самым взаимное трение по-
токов, что в свою очередь мешает образованию кон-
вективного круга и приводит к уменьшению скорости
тепловой конвекции.

Также при углах наклона α = −90◦ и α = 90◦ заме-
тен спад скорости конвекции. Объясняется это тем,
что при нагреве области горизонтально расположен-
ной пробирки тепловые потоки ударяются о «стенку» и
приводят к снижению скорости конвекции таким обра-
зом, что потоки просто вытесняют друг друга. В случае
с α = 90◦ подъемное нагревательное течение сверху
ограничено, что вынуждает жидкость течь вдоль стен-
ки постепенно охлаждаясь. Здесь учитываются силы
трения о «стенку» и условие прилипания, замедляющее
такой поток. То же справедливо и про охлаждающий
поток. При α = −90◦ обеспечивается более подходящее
условие для всплытия нагретой жидкости и опускания
охлажденной при минимальном взаимном трении и
трении со «стенкой», что приводит к чуть более высокой
скорости конвекции. Соответственно при таких углах
наклона скорость конвекции не самая эффективная.

7. Заключение
В настоящей работе рассмотрена ранее смодели-

рованная модель миркопробирки в условиях темпера-
турной конвекции. Произведены расчеты с использова-
нием встроенного решателя, а также их постобработка
с последующим применением средств языка програм-
мирования Python. По результатам были проведены
анализ полученных результатов и оценка качества пе-
ремешивания в зависимости от угла наклона пробирки.

Важно отметить, что моделирование температур-
ной конвекции в микропробирках представляет собой

важнуюобластьисследований сширокимспектромпри-
ложений, начиная от микрофлюидики и биомедицин-
ских технологий до электроники и материаловедения.
Оптимизация и дальнейшее исследование рассматри-
ваемых процессов актуально уже более двух столетий.

По результатам исследования выявлено существен-
ное влияние угла наклона на результаты, что говорит о
том, что предположение, выдвинутое в начале статьи,
было верным.

Результаты исследования позволяют лучше понять
влияние угла наклона пробирки на скорость конвекции
жидкости, что имеет значение для оптимизации про-
цессов, связанных с теплообменом и перемешиванием
в микромасштабных системах. Полученные данные мо-
гут быть использованы для разработки новых методов
контроля и управления конвекцией в микропробирках,
что открывает перспективы для создания более эффек-
тивных и точных микросистем.
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1. Introduction
All currently known technologies of the production of

high-temperature composite materials are being improved
constantly and, due to their increasing complexity, require
the implementation ofmore perfect approaches at different
stages [1–3], including the very first one— the construction
of new physical and mathematical models for the process
occurring. One of the current problems in this area is the
numerical simulation of the gaseous silicon transfer from
the melt surface to the product during high-temperature
siliconizing of a carbon porous material. For real produc-
tion conditions, the complex system of differential equa-
tions in partial derivatives must describe this process. The
adequate modelling of this process requires taking into
account many complicating factors, including compress-
ibility of the medium. At the same time, the technology
is essentially three-dimensional and requires a highly de-
tailed computational mesh due to the presence of many
crucibles with the melt in retort and their complex loca-
tion in the working space of the furnace. As the result, it
is not yet possible for a technologist to carry out quickly
full-fledged 3D numerical simulation of this process. At
the moment, existing models for describing the transfer of

gaseous silicon in a working space of the industrial instal-
lation during high-temperature siliconizing are limited to
the simplest approaches [4]. Namely, due to the fact that
this process is carried out under conditions of moderate
vacuum and at extremely high temperatures above themelt-
ing point of silicon, it is believed that the diffusion plays
decisive role in ensuring the transport of gaseous silicon
from the melt surface to the product [5]. In other words,
to simulate this process, the simplest diffusion equation is
always used without taking into account additional mecha-
nisms, like convective transport. At the same time, it turns
out that the use of real values of the diffusion coefficient
in this equation does not allow us to ensure the supply of
silicon amount observed in the experiment, which is neces-
sary for complete siliconizing of the product. As the result,
there is a paradoxical situation, because the facts speak for
themselves: in experiments, under certain conditions, the
product can still be saturated with the required amount of
silicon but calculations based on diffusivemodel predict im-
possibility of this process. This means that all the physical
conditions, necessary for the successful implementation of
this process, still remain not fully understood.

http://mfs.uimech.org/mfs2024.3.015
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2. Diffusive and Convective Transport in
Gases

2.1. Statement of the problem

At first, let us ignore the details of real industrial tech-
nology and consider the process in the simplest model for-
mulation, when the surfaces of the product and the silicon
melt represent two parallel planes located at a distance L
from each other (Figure 1,a). The working space of retort
is filled up with a residual gas, the role of which the argon
plays. The conditions of medium vacuum are maintained
in the retort and correspond to the partial pressure about
100 Pa [6,7]. The temperature exceeds the melting point of
silicon. The product 1 is saturated by silicon vapour which
is generated by the melt 2 (Figure 1,a). There is a task to
study the transfer of silicon vapor from the melt surface to
the product through residual gas. Characteristic size of the
retort accounts for about 1–2 m, therefore in our statement
we can neglect the effect of gravity (quantitative estimation
of this effect will be given in section 2.2). At first, let us
begin the discussion about phenomenon description from
the analysis of the classical diffusion equation, which has
in the three-dimensional case the following form

∂C
∂t

= D
(

∂2C
∂x2 + ∂2C

∂y2 + ∂2C
∂z2

)
, (1)

where D is the diffusion coefficient, C is the mass concen-
tration of gaseous silicon. This is a standard second order
partial differential equation of parabolic type. In the sta-
tionary case (∂/∂t = 0), the problem (1) is simplified and
reduced to the Laplace’s equation ∆C = 0.

The non-applicability of this equation for the simula-
tion of silicon vapor transfer becomes obvious in a limiting
case of one-dimensional statement (∂/∂y = ∂/∂z = 0).
For the simplest homogeneous boundary conditions

C(0) = 0, C(L) = Cs, (2)

only linear distribution of silicon concentration satisfies to

the one-dimensional Laplace’s equation

C(x) = Cs

L
x. (3)

HereCs is the concentration of silicon vapor on themelt sur-
face. Obviously, this value can’t exceed the concentration
of saturation for given thermodynamic conditions. The zero
value of concentration on the left boundary means com-
plete absorption of gaseous silicon by the porous medium.
Characteristic distance L between the melt surface and the
product in the course of the technological process is equal
about 0.5–1 m. The saturated vapor pressure for silicon at
the temperature not much higher than the melting point
of silicon is very small and accounts in order of magnitude
to ps = 10 Pa [8,9]. Thus, the number of silicon atoms per
unit of volume for conditions of saturation is estimated
using the state equation for the ideal gas

ns = ps

kT
= 10

1.38 · 10−23 · 1800
= 4.03 · 1020 1/m3, (4)

where k is the Boltzmann’s constant. The characteristic
working temperatures lie near T = 1800−1900 K. It is im-
portant to compare the theoretically predicted silicon flux
density with the weight gain of the product observed in
experiment. In fact, by definition the flux density of a sub-
stance is the mass transferred per a unit time through a
unit area. If silicon transport is determined only by the
diffusion, then in this case the Fick’s law describes silicon
flux density

jSi = −ρD∇C = −D∇ρSi. (5)

Let us make the estimation of diffusion coeffi-
cient using the well-proven formula of molecular-kinetic
theory [10]:

D = 3
8

kT
σ12 p

√
πkT
2µ12

= 3
8

(kT)3/2

d2
Sip

√
πm0

. (6)

Here σ12 is the effective scattering cross-section for two par-
ticles, µ12 is the equivalent mass. For two approximately
equal masses and sizes for silicon atom and argon one, we

Figure 1. a) Plane geometry of the problem: 1 — product surface (carbon material), 2 — surface of silicon melt (melt mirror), 3 — linear
distribution according to the standard diffusion model, 4 — working space of retort; b) Analytical solution of modified diffusion
equation for L = 0.5 m
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apply σ12 ≈ πd2, µ12 ≈ m0/2; where m0 is the mass of one
silicon atom m0 = MSi/NA = 0.028 kg/Mole / 6.02 · 1023

1/Mole = 4.7 · 10−26 kg (MSi is the silicon molar mass). In
addition, the diameter of this atom accounts about dSi =
0.23 · 10−9 m. The resulting value of the diffusion coeffi-
cient is D = 0.7 m2/s. Despite of so giant and unusual
value of diffusion coefficient, the formula (5) for constant
gradient Cs/L predicts a very low value of silicon flux den-
sity jSi = 2.62 · 10−5 kg/m2 s (silicon density on the melt
mirror is ρSi = pSiMSi /RT = 1.87 · 10−5 kg/m3, where R is
the universal gas constant).

According to the estimations of technologists involved
in the siliconizing of carbon products, this is clearly not
enough to block the pores completely in a reasonable time
for existing porosity of the material. Moreover, today there
is a situation when the moving to a new workpiece shape
or new heating conditions, the process cannot be predicted
whether it will be successful. However, in reality, if certain
conditions are found experimentally, the product of any
shape is still successfully saturated with silicon. Thus, an
assertion can be reliably established according to which all
failures during the technological process are determined
not by the diffusive resistance of residual gas and the large
distance from the melt surface to the product, but by ab-
solutely another factors.

2.2. Convective transport in residual gas

Thus, it is necessary to explain the anomalously in-
tensive transport of gaseous silicon from the melt to the
product surface, which takes place in experiments. The
more general equation of admixture transfer in continuous
media mechanics, taking into account an additional con-
vective mechanism of the transport [11], is written in the
form

∂C
∂t

+ (V∇)C = D∆C. (7)

Here V is the macroscopic mean-mass velocity of a physi-
cally small fluid element. The main problem of the equa-
tion usage is in the absence of mathematical completeness,
since the velocity field is an additional unknown quantity.
It seems that convective mechanism is the next significant
factor after diffusion in the course of silicon mass transfer.

Within the framework of continuous media mechan-
ics, the dynamics of fluid is determined generally by the
Navier–Stokes equation [11]. In three-dimensional case,
there are three evolutionary nonlinear partial differential
equations for velocity components Vx, Vy and Vz. These
equations include in addition two variables — the pressure
and the density, which need to be determined also. As the
result, two supplementary equations are added to the sys-
tem: the lawofmass conservation for compressiblemedium
in differential form and the equation of state. Thus, the
final system of equations becomes extremely cumbersome
and difficult to solve.

Currently, direct numerical simulation of the pro-
cesses under consideration in a full three-dimensional
formulation is very difficult, even with the help of high-
performance supercomputers. The task is to reformulate

the problem more simply. On the one hand, it is impor-
tant to take into account all the physical factors necessary
for the adequate description of the process under consid-
eration, and on the other hand, to avoid the use of overly
complex models. It is desirable to construct a new phys-
ical model that can permit to carry out our calculations
in a reasonable time and would not require an excessive
computing power.

To realize these perspectives let us follow the approach,
which is implemented in the theory of porous media. Resid-
ual gas is not generated at the melt surface and is not ab-
sorbed by the product. It permits to suppose that the trans-
fer of silicon vapor through the residual gas is analogous
to filtration of a fluid through porous medium, and this
average flow is not reduced to the diffusion phenomenon.
Then, let us take the momentum equation for filtration in
its primary formulation (Wooding, 1957) and evaluate the
contribution of each term in it. In the hydrodynamics of
porous media, it is necessary to distinguish between the
intrinsic velocity v and the seepage velocity u. The seepage
velocity is determined through the total fluid flow rate and
is related to the pore velocity by the ratio u = ϕv. Here ϕ
is the material porosity. For intrinsic velocity in the porous
medium mechanics, we can use the equation of the fluid
motion in first rough approximation [12], neglecting by
Brinkman’s and Vafai’s comlexifications

ρ f

(
∂v
∂t

+ (v∇)v
)

= −∇p − η

κ
u + ρ f g. (8)

Here ρ f is the density of the pumped fluid, η is the dynamic
viscosity, κ is the permeability. Turning completely to the
seepage velocity, we get the equation in the form of Dupuit-
Forchheimer

ρ f

(
ϕ

−1 ∂u
∂t

+ ϕ
−2(u∇)u

)
= −∇p − η

κ
u + ρ f g. (9)

Now it is needful to evaluate the terms containing ve-
locity in the left and right parts of this equation. The most
non-trivial parameter in this equation is permeability. In
relation to the mobile atoms of the residual gas (argon), we
can only talk about the model nature of this matter as a
porous medium with some effective permeability. Namely,
we mean a model according to which the gaseous silicon,
as a kind of fluid, is filtered through a carrier medium due
to the fact that silicon vapors are scattered on dotty argon
atoms, which heavier than silicon atoms, but neverthe-
less mobile. Due to the extreme rarefaction of the carrier
medium, it is expected that the permeability has anoma-
lously large value.

Indeed, the value of mean free path corresponding to
the conditions of moderate vacuum is equal

l = kT√
2πd2 p

= 1.38 · 10−23 · 1800√
2(1.4 · 10−10)2100

= 2.8 · 10−3 m. (10)

The value of mean free path permits to calculate the
Knudsen number. It plays key role in the mechanics of
rare gases. The distance between the product and the melt
mirror determines characteristic spatial scale. In our case
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L ∼ 1 m. Following to definition, the Knudsen number can
be estimated as

Kn = l
L

∼ l = 2.8 · 10−3 m.

One can conclude that Knudsen number satisfies con-
dition Kn ≪ 1, that leads to the ability of continuum me-
dia mechanics application. On the other hand, this value
close to the limits of applicability of this approach. This
is the actual reason of the paradoxical situation that de-
mands taking into accounts of additional factors to describe
anomalously intensive transfer of silicon vapour.

Nevertheless, let us apply the Kozeny–Karman for-
mula to estimate the permeability of this artificial porous
medium. For partial pressure of argon 100 Pa and porosity
ϕ = 1 − v0nAr = 0.9999999979, the permeability has the
value

κ = κ0ϕ
3

(1 − ϕ)2 = 2 · 10−3 m2, (11)

where v0 is the volume of one argon atom, κ0 is the
Kozeny–Karman constant. It has supposed in our calcu-
lation that following estimation is valid for this parameter
κ0 ∼ a2; where a is the characteristic size of the obstacle
(for spherical argon atom a ∼ 10−10 m).

Another important quantity is the macroscopic mean-
mass velocity of the gas element V. We have to distinguish
later among three velocities: the mean-mass velocity V,
which appears in equation (7), the seepage velocity u and
the intrinsic one v, that enter into equation (8). Note that
the values of seepage velocity and intrinsic one are approx-
imately equal to each other because our medium is rare and
its porosity is extremely close to unit. It can be assumed for
estimation that during evaporation, silicon atoms detach
from the melt surface with a root-mean-square speed

u =
√

3kT/m0 ≈ 1250 m/s. (12)

An averaging this velocity over all directions gives the
mean value of the silicon flux density out of the melt in
projection on the normal jSi = ρSiu/2. The definite part of
the vapor momentum is transmitted to the scattering cen-
ters therefore the equality for themean-mass velocity takes
place ρSiu/2 = (ρSi + ρAr)V. As the result, the mean-mass
velocity is defined by the formula V = ρSiu/2(ρSi + ρAr) =
41 m/s (for partial pressure of argon pAr = 100 Pa the den-
sity is equal to ρAr = ρArMAr/RT = 2.67 · 10−4 kg/m3). This
evaluation means that convective mechanism makes the
significant contribution to the general transfer of gaseous
silicon during the process under consideration.

Let us estimate now the value of each term in equation
(9), taking into account that the porosity of such medium
is close to unit. In our statement, the flux and the seepage
velocity of gaseous silicon have only one component along
x-axis, which can depend on all three coordinates, there-
fore v∇v ̸= 0 for compressible fluid. On the other hand, the
conditions for steady transfer are settled extremely quickly.
Thus, we can assume ∂v/∂t = 0 for steady regime of trans-
fer. Thus, the non-linear term, viscous one and gravity

force in (9) are equal in order of magnitude to following
values∣∣∣ρ fϕ

−2(u∇)u
∣∣∣ ∼ ρ f

u2

L
= 1.87 · 10−5 6252

1.0
= 7.3 ,∣∣∣ηu

κ

∣∣∣ = 3.5 · 10−4 · 625
2 · 10−3 = 110 ,∣∣∣ρ f g

∣∣∣ = 1.87 · 10−5 = 1.87 · 10−4

(13)

and satisfy the condition∣∣∣ηu
κ

∣∣∣ ≫
∣∣∣ρ fϕ

−2(u∇)u
∣∣∣ ≫

∣∣∣ρ f g
∣∣∣ . (14)

One can see that there is no parity between these three
terms. Comparison of these quantities permits to neglect
by the inertial term and gravity influence in equation (9).
As the result, the Darcy’s law follows from this equation

u = −κ

η
∇pSi. (15)

Due to the incessant evaporation on the melt surface
and absorption on the product, we have a certain average
density gradient of silicon vapor. Since the pressure in the
ideal gas is proportional to density, this causes a silicon
pressure gradient, which can act as an additional driving
force differing from diffusion. The partial pressure of sili-
con according to the equation of the ideal gas state is equal
to

pSi = nSikT, (16)

where nSi = NSi/V — number of silicon atoms per unit
volume. Let us express nSi over the mass concentration C.
By definition, we introduce mass concentration over the
formula

C = ρSi
ρAr + ρSi

. (17)

Silicon density is expressed from (17) in terms of rela-
tive mass concentration as follows

ρSi = C
1 − C

ρAr. (18)

Let’s rewrite equation (16) in terms of silicon density,
i.e. substitute relation (18) into (16) at second step:

pSi = RT
MSi

C
1 − C

ρAr. (19)

Then, we substitute this result into Darcy’s law (15),
for simplicity neglecting the spatial inhomogeneities of
argon density and temperature. Let’s also take into account
the fact that the silicon mass concentration never actually
reaches the unit. Argon, as a residual gas, always presents
in the retort, and their concentration is approximately an
order of magnitude higher than the concentration of silicon
vapor. As the result, we expand the factor C/(1 − C) into a
series in small C and limit ourselves in the final expression
to the first non-vanishing term. Darcy’s law (15) takes the
final form

uSi = −κ

η
∇pSi = −κ

η

RTρAr
MSi

∇C. (20)
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However, it is necessary to substitute the average-mass
velocity into equation (7), which is calculated as

V = ρAruAr + ρSiuSi
ρAr + ρSi

= ρSiuSi
ρAr + ρSi

≈ ρSiuSi
ρAr

. (21)

In this approximation the argon is immovable uAr =
0. According to our model, the silicon vapour is filtered
through the stationary argon. Substituting this expression
into the generalized equation (7) for impurity transfer, we
finally obtain the equation

∂C
∂t

− κRTρSi
ηMSi

(∇C)2 = D∆C. (22)

Now this is a more complex partial differential equa-
tion of diffusionwith nonlinearity like the square of the con-
centration gradient. Note that similar diffusion equations
with nonlinearities that are quadratic to the concentra-
tion gradient are quite common in various fields of physics.
Thus, in works [13, 14] it is shown that a nonlinear term
of this type quite significantly changes the transport diffu-
sion properties of the material, which is played by a lithium
niobate crystal, and makes it possible to explain some of
the observed effects associated with the saturation with hy-
drogen of the material under consideration. For stationary
conditions, equation (22) is reduced to the form

−(∇C)2 = ψ∆C, (23)

where ψ is a positively definite, dimensionless parameter
responsible for the contribution of the convective term to
the overall transport of silicon vapor

ψ = ηMSiD
κRTρSi

. (24)

It is important to estimate the value of the new intro-
duced parameter. Let us use the values of dynamic viscosity
and diffusion coefficient as in the works [6,7]. These param-
eters were assessed in relation to our siliconizing process
with the help of the well-known formulas of the molecu-
lar kinetic theory [10]. There is the formula for dynamic
viscosity for a rare gas

η = 1
3

(
2
π

)3/2
(m0kT)1/2

d2 . (25)

Here d is the diameter of atom; m0 is the atom’s mass. For-
mula (25) gives the same magnitude of dynamic viscosity:
η = 3.5 · 10−4 m2/s. Thus, according to (6), (11), (25) pa-
rameter ψ quantitatively has the value

ψ = 3.5 · 10−4 · 28 · 10−3 · 0.7
2 · 10−3 · 8.31 · 1800 · 1.87 · 10−5 = 0.012. (26)

Note that this complex is an initially dimensionless
parameter in the problem (23) and characterizes the rela-
tionship between diffusive and convective transport. The
smallness of this parameter in (23) means that the new non-
linear term in the left part of the equation has an essential
role in our model.

Of cause, the parameter (24) in most cases is greater.
This parameter is small only for conditions of medium vac-
uum during product siliconizing in the course of techno-
logical process. At normal conditions, the density of car-
rying gas becomes significantly bigger and this leads to
the diminution of permeability. It is seen from formula
(24) that this parameter at lower temperature becomes also
greater. As the result, the parameter ψ increases and we
have purely diffusive transfer.

3. Exact Solutions of Equation
for Concentration

3.1. Plane one-dimensional geometry

Thus, we have only one closed equation (22) for silicon
vapor transfer with well-known parameters. Owing to the
comparative simplicity of this equation, the pleasant ability
exists to find an exact analytical solutions in several actual
statements. The first one is one-dimensional stationary
distribution of silicon vapor in Cartesian coordinate system
with boundary conditions of the first kind. The equation
(23) in this case has the form

−
(

∂C
∂x

)2

= ψ
∂2C
∂x2 . (27)

It is supplemented with the Dirichlet boundary con-
ditions

C(0) = 0, C(L) = Cs. (28)

Using procedure the change of variable [15], the inte-
gration of equation of (27) with boundary conditions (28)
gives the solution

C(x) = ψ ln

(
x
L

(exp(Cs/ψ) − 1) + 1

)
. (29)

The functional dependence (29) is presented in Figure
1b for Cs = 0.08 and ψ = 0.012. This dependence qualita-
tively different from the linear solution (3). It means that
the silicon vapor tends to fill up completely the working
space of retort. Derivative of (29) on the product surface
is determined by formula

∂C
∂x

∣∣∣
x=0

= ψ(exp(Cs/ψ) − 1)
L

. (30)

It can be seen that this derivative is significantly
greater than the same one for the linear solution. Thus,
one can explain the anomalously intensive absorption of
gaseous silicon by the product during silicification. In the
limiting case of small ratio Cs/ψ the exponent can be ex-
panded into series and we have got the derivative for linear
solution (3), which is equal to Cs/L.
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3.2. Cylindrical one-dimensional case
Analogical problem can be solved in cylindrical coordi-

nate system. Using conditions of concentration uniformity
on azimuthal angle and axial coordinate, we have the equa-
tion

−
(

∂C
∂r

)2

= ψ

r
∂

∂r

(
r

∂C
∂r

)
. (31)

As before, the Dirichlet boundary conditions can be
imposed in our case

C(R1) = 0, C(R2) = Cs. (32)

This boundary value problem describes steady transfer
of silicon vapor form external uniform sources located at
distance R2 − R1 to the product of cylindrical form with
radius R1 (Figure 2,a).

As the result of statement with cylindrical symmetry,
the integration of equation (31) for boundary conditions
(32) gives superlogarithmic solution

C(r) = ψ ln
(

(ln(r/R2) − exp(Cs/ψ) ln(r/R1)
ln(R2/R1)

)
. (33)

Derivative of (33) at r = R1 is determined by expres-
sion

∂C
∂r

∣∣∣
z=R1

= ψ

R1

(1 − exp(Cs/ψ))
(ln(R1/R2) − exp(Cs/ψ)) . (34)

It is seen that in the case of modified equation (22) the
values of derivative as for rectangular coordinate system
as for cylindrical one are evidently greater than analogical
one for classical diffusion equation (1). The distinction can
exceed the whole order. Note that the mass flux is propor-
tional to the derivative of concentration. Thus, it means
that the vapor-liquid phase method could be soundly ap-
plied for silicification of carbon porous materials and our
theory confirms ability of the technology realization.

Figure 2. a) Cylindrical geometry of the problem with vertical
melt mirror: 1 – product surface, 2 – melt mirror, R2,
R1 – radii of cylindrical shells, Cs – concentration of
saturation; b) Cylindrical geometry with horizontal melt
mirror

4. Numerical Simulation in Two-
dimensional Statement

4.1. Two-dimensional problem with the axial
symmetry

The problem of silicon vapor transfer looksmore realis-
tic for the systems with cylindrical symmetry. The simplest
configurations are presented in Figure 2, where one can see
two coaxial cylindrical shells with different radii R2 and R1,
corresponding to the melt surface and cylindrical product.
Now let us consider the general case of time-dependent
statement to observe also the formation and motion of
concentration front. On the other hand, we simplify the
statement assuming the condition of axial symmetry of
the system. Thus, the equation (22) for unknown function
C(r, z, t) in cylindrical coordinates can be written as

∂C
∂t

− Dc

(∂C
∂r

)2

+
(

∂C
∂z

)2
 =

= D

(
∂2C
∂r2 + 1

r
∂C
∂r

+ ∂2C
∂z2

)
.

(35)

Here Dc is the dimensional parameter, which character-
izes the intensity of convective transfer in comparison with
diffusion. Note, that there is no necessity to introduce
non-dimensional quantities in our problem because this
parameter has the same dimension [m2/s] as diffusion coef-
ficient. There is a mathematical ability to compare Dc and
D. By definition, this new parameter is equal to

Dc = κRTρSi
ηMSi

. (36)

Further, we shall call it as the parameter of convec-
tive transport, which appears in physics for the first time.
For our conditions it has the value Dc ≈ 58 m2/s (see esti-
mation ψ). Besides of the sectors with sources and sinks
the impermeable boundaries of retort appear in this state-
ment. Thus, two sets of boundary conditions take place for
corresponding configurations, which are shown in Figure 2:

1) C(R1, z) = 0, C(R2, z) = C,
∂C
∂z

∣∣∣
z=0

= ∂C
∂z

∣∣∣
z=H

= 0.

2) C(R1, z) = 0,
∂C
∂r

∣∣∣
z=R2

= 0,
∂C
∂z

∣∣∣
z=H

= 0;

R1 ⩽ r ⩽ Rs :
∂C
∂z

∣∣∣
z=0

= 0,

Rs < r ⩽ R2 : C(r, z = 0) = Cs.

(37)

According to the first statement, the surfaces of prod-
uct and melt mirror are opposite to each other. Under the
second set of boundary conditions, the surface of the melt
mirror is horizontal and surrounds the vertical product shell
like in an experiment.
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4.2. The method of numerical simulation

Nowwe have an evolutionary differential equation (35)
in partial derivatives, which is solved numerically by the
standard method of finite differences [16]. It is important
to use the method that can be applied in a future for autom-
atization of technological process. The method of finite
differences is the most suitable for this purpose realiza-
tion. An explicit scheme of first-order accuracy by time
is applied to simulate the evolution of the system. In cal-
culation algorithm, the spatial derivatives have different
accuracy to provide the stability of the scheme. We use cen-
tral second-order accuracy formulas for derivatives approx-
imation along axial coordinate z and first-order accuracy
backward differences along radial coordinate r. The solu-
tion of equation (35) with boundary conditions (37) is found
for reasonable values of dimensional parametersH = 0.4m,
L = R2 = 0.9 m, R1 = 0.2 m, D = 0.7 m2/s, Dc = 58 m2/s
(L is the full retort radius, which can distinguish from the
radius of the shell with the melt). The time step is chosen
in dependence on spatial one to obtain the stability of the
numerical procedure (the Courant criterion). The author’s
computer code has implemented in the FORTRAN-90 pro-
gramming language. The space of retort is initially empty,
which means that at the beginning the concentration of
the silicon vapor is equal zero in calculation domain. There
is used a numerical grid with the number of nodes along
radial and axial coordinates 85(121):41; along radial axis
85 nods – for vertical melt surface (Figure 2,a), 121 — for
horizontal one (Figure 2,b). The ratio of the retort radius L
to the product height H has been varied in the interval 2–4.

4.3. Numerical results and discussion

Two-dimensional numerical calculations show some
complementary features of the process. This computation
permits to ascertain the characteristic time of the settling
of silicon vapor distribution in retort. The fields of silicon
concentration in section (r,z) are presented in Figure 3 for
two moments of time in the case when the surfaces of melt
and product are opposite (Figure 2,a). Following values of

time points correspond to these realizations: t1 = 0.008 s,
t2 = 0.5 s.

At initial stage the concentration front of silicon va-
por originates on the external boundary and moves inside
to the product surface. It is visible that the gaseous sili-
con propagate quickly through the entire space of retort.
One can see that the settling of stationary concentration
profile occurs approximately over the time t = 0.5 s. At
the final steady stage, the diminution of concentration is
only observed near the product surface, which absorbs the
silicon vapor according to the boundary condition. The
profile on the Figure 3,b looks like the analytical solution
(33). Also it corresponds qualitatively to the solution with
plane geometry (29) which is shown in Figure 1b,.

It must be emphasized that in the case of cylindrical
symmetry, one can observe the more strong effect of wave
front steeping in comparison with the plane statement. In
the system with cylindrical symmetry the streamlines are
concentrated during the motion of gaseous silicon from
the periphery to the centerline therefore the additional in-
creasing of flux density on the surface product is observed
owing to the topological convergence effect.

Then, let us consider the problem with more realistic
statement, when the mirror of silicon melt is horizontal
and the product surface, as early, is vertical (Figure 2,b).
The width of the melt ring is equal ∆R = L/3. The fields of
silicon concentration are shown in Figure 4. At first stage
the gaseous silicon becomes to evaporate vertically upward
(Figure 4,a). After that the streamlines turn to the product
surface and the silicon vapor begins again to fill up quickly
the working space of the retort.

It can be concluded that the filling of the working space
in the retort by the atoms of gaseous silicon occurs after the
beginning of evaporation less than for the second. Mainly
due to the convective transfer, the concentration of silicon
in the volume quickly amounts to the value, which is close
to the one near the melt surface. If conditions of silicon
condensation take place on the product surface, the high
gradient of concentration originates at this boundary after

Figure 3. The concentration of silicon vapor in vertical section for the opposite surfaces of the melt and product for two moments of
time; a) — t1, b) — t2
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Figure 4. The concentration of silicon vapor in vertical section for the horizontal surface of the melt mirror for two moments of time a) —
t1 = 0.005 s, b) — t2 = 0.1 s

the settling of steady state. The pressure flow is realized
in this system because of the intensive evaporation at the
melt surface. So, not only diffusion causes silicon transport
through the residual gas but also convective transfer.

Theoretical works [4,5] predicted the impossibility of
silicon transfer in sufficient quantities for the realistic time.
In the result of solution of the classical diffusion equation,
the estimation of the full siliconizing time is equal 20–30
hours [4,5]. The time of the siliconizing is proportional to
the derivative of concentration at the surface of the product,
see formulas (30) and (34). Our calculations demonstrate
that the concentration derivative at the product bound-
ary is ten times greater than in the case of pure diffusion.
Thus, we prognosticate the siliconizing time 2–3 hours that
conforms experimental data very well.

5. Conclusion

In the results of analytical and numerical calculations,
one can summarize that there is no necessity to approach
the crucibles as possible to the product during technolog-
ical process in order to overpass the diffusive barrier of
residual gas. This point of view could be valid only for lin-
ear distribution of gaseous silicon concentration in retort.
Indeed, in this case the technologist have to move crucibles
closer to the product in order to increase as possible the
concentration derivative on the sample surface.

It has proved analytically, and additional numerical
calculations confirm this result, that the convective trans-
fer of gaseous silicon makes considerable contribution into
the full mass flux in real technological conditions.

It is shown theoretically, that silicon vapor tends to
occupy quickly the whole working space in retort therefore
near the product surface there is always sufficient amount
of silicon for successful realization of the siliconizing pro-
cess. The simplicity of numerical procedure realization con-
firm an efficiency of suggested new non-linear diffusion
equation for description of silicon transfer in the conditions
of medium vacuum.
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Numerical modelling of the effect of gas temperature non-uniformity
on the geometric parameters of the heating element
in a cold spraying technology
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In contemporary mechanical engineering, extending the lifespan of products necessitates heightened standards for the materials used in the
fabrication of components and structures. One of the most promising avenues is enhancing material characteristics through the application
of functional coatings. This includes boosting the material’s corrosion resistance, wear resistance, and protection against mechanical
damage, as well as enabling localized repairs without disassembling the structure. Preference is given to technologies that do not adversely
affect the surface to which they are applied. Cold spray technology stands out as one of the most rapidly advancing methods for applying
protective coatings and imbuing materials with various functional properties. This technique not only safeguards surfaces but also enhances
their operational characteristics, ensuring the longevity and reliability of products. Investigation of the influence of the dependence of
the gas heating on the geometric parameters of the heating element of the cold spraying technology was considering in this work. This
task was considered in two software solvers: the software product ”Thermoviscous fluids: a hydrodynamic simulator for modeling flow in
annular channels with heat exchange“ and ANSYS. The modeling results show that the spiral steel tube can effectively be used for heating
gas to high temperatures at high speeds. However, it is necessary to consider that at high speeds, additional hydrodynamic effects such
as turbulence and shear flows may occur, which can affect the efficiency and stability of the gas (nitrogen) flow.

Keywords: gas heating, heating element, cold spraying technology, nitrogen

The research work was supported by the state budget funds for the state assignment 124030400064-2 (FMRS-2024-0001).

1. Introduction
In the realm of modern mechanical engineering, ex-

tending the service life of products demands elevated stan-
dards for thematerials employed in themanufacture of com-
ponents and structures [1–3]. One of the most promising
approaches is to enhance material characteristics through
the application of functional coatings. This includes aug-
menting the material’s corrosion resistance, wear resis-
tance, and protection against mechanical damage, as well
as facilitating localized repairs without disassembling the
structure [4]. Technologies that do not negatively impact
the applied surface are favored.

Cold spraying (CS) technology is at the forefront of
methods for applying protective coatings and endowing
materials with various functional properties [5, 6]. This
process, broadly known as cold spraying, aims to develop
methods and devices for spraying coatings that impart the
necessary properties to product surfaces, such as reduc-
ing porosity, increasing hardness, strength, and corrosion
resistance.

The primary objective of these innovations is to cre-
ate a method and device for spraying coatings that elevate

the surface properties of products to the required levels,
ensuring reduced porosity, enhanced hardness, increased
strength, and superior corrosion resistance. Over the past
20 years, the greatest inventive activity in the field of CS
coating technologies has been observed in Russia, Germany,
China and the USA [7]. Also, this coating area is also being
developed by developers from Japan, the Republic of Korea,
India, Austria and Australia [8].

The forefront of CS technology development is marked
by a drive to enhance productivity and automate the coating
application process. This is achieved through the creation
of novel automated systems and the exploration of modern
powder materials, which endow parts and products with
diverse functional properties.

The device for applying coatings via the CS method
comprises several key components: a dosing feeder, a hous-
ing that includes a powder hopper shaped like a drum with
recesses on its cylindrical surface, and a mixing chamber
equipped with a nozzle designed to accelerate powder par-
ticles. The nozzle is connected to both the mixing cham-
ber and a compressed gas source. A powder particle feed
controller ensures the desired flow rate during the coat-
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ing process, while a partition installed at the bottom of
the hopper prevents powder particles from entering the
space between the drum and the dosing device body, thus
avoiding potential jamming.

The nozzle, featuring a profiled passage, enables the
gas flow to reach supersonic speeds. To further boost pro-
cess efficiency and regulate the speed of the gas-powder
mixture with a supersonic jet, the device incorporates an
element for heating compressed gas, complete with a tem-
perature control system.

Preferably, the heating element should be made from a
resistor alloy, which will allow for a reduction in the overall
dimensions and weight of the heating means [9–13].

In the work ofWenya Li [14], an experiment was carried
out and a 2D model of the entire process of CS technology
was constructed, however, numerical modeling of the gas
heating element was not performed. Nitrogen is most often
used as a working gas.

Nitrogen is an inert diatomic gas without color and
odor, the chemical formula of the diatomic molecule N2,
molar mass 28.01 kg/kmol. The nitrogen content in the
atmospheric air is about 78.09 % by volume. It is used in
technological processes as an inert sealing medium, for
example, for dry gas mechanical seals and sealing com-
plexes, in the chemical industry for the synthesis of ammo-
nia. Liquid nitrogen is used as a refrigerant in mechanical
engineering for the assembly of non-removable tight joints
(cooling of the covered part). Nitrogen is used in special
technological processes for applying a thin layer of wear–
resistant coating — titanium nitride on the surface of steel
parts; in combination with silicon, it forms a wear-resistant
promising ceramic material, silicon nitride Si3N4 [15].

Numerical simulation of the gas heating process will
allow to set the optimal parameters of the heated element
to reduce the gas heating time, and, accordingly, optimize
the process in the complex.

2. Problem statement
The working gas is nitrogen (N2), which is supplied to

the tube at room temperature, 293 — 298 K (20 — 25 ◦C),
and a pressure of 2 to 10 MPa. The tube walls are main-
tained at a constant temperature of 1273 K (1000 ◦C). The
inner diameter of the tube is fixed and is 0.008 m, and the

wall thickness is 0.002 m. The diameter, number of coils,
and length of the tube are parameters that need to be de-
termined for efficient gas heating.

3. Methods
This task was considered in two software solvers:

1. The software product «Thermoviscous fluids: a hy-
drodynamic simulator for modeling flow in annular
channels with heat exchange» [16].

2. ANSYS — a universal finite element analysis (FEA)
software system

4. Numerical results
4.1. Part 1. Results of numerical simulation in the

own software product

The flow of incompressible nitrogen in a flat channel
was considered. The channel diameter was fixed, and the
channel length L varied from 1 m to 50 m. The modeling
environment used was the own software product «Ther-
moviscous fluids: a hydrodynamic simulator for modeling
flow in annular channels with heat exchange» which was
adapted to solve this problem: the inner diameter of the
tube — 0.008 m; the length of the tube — from 1 m to 50 m;
the pressure drop dp — 1 Pa; the initial temperature of the
gas — 293 K; the temperature of the tube walls — 1273 K.

To solve the problem of the incompressible flow of ni-
trogen in a flat channel, the control volume method was
chosen. This approach allows solving the Navier–Stokes
equations with high accuracy, which is critical for obtain-
ing reliable results.

The geometric dimensions of the tube, the properties
of nitrogen, the boundary conditions for velocity and pres-
sure at the inlet and outlet of the tube, and the temperature
at the tube walls were used as input data for modeling.

Based on the calculations, the results for the distribu-
tion of velocity, pressure, and temperature along the length
of the tube were obtained. The results of the test calcula-
tion for a channel with a length of 1 m and a pressure drop
of 1 Pa are shown in Fig. 1. The graph shows the distribu-
tion of temperature and velocity along the length of the
channel in a dimensionless form.

Figure 1. Distribution of temperature and velocity in the computational domain for L = 1 m at dp = 1 Pa
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Figure 2. Distribution of temperature and velocity in the computational domain for L = 10 m at dp = 104 Pa

Figure 3. Distribution of temperature and velocity in the computational domain for L = 50 m at dp = 105 Pa

From the obtained results, it follows that the distribu-
tion of the nitrogen velocity along the length of the channel
has a parabolic shape. The maximum velocity is reached
in the middle of the channel and decreases to zero at the
tube walls.

The study results showed (Fig. 2) that as the chan-
nel length increases to 10 meters and the pressure drop
to 104 Pa (equivalent to 0.1 atm), the nitrogen is heated
closer to themiddle of the channel. Themaximumnitrogen
velocity in the channel is 0.7 m/s.

Further, when the channel length is increased to 50me-
ters and the pressure drop to 105 Pa (equivalent to 1 atm),
the nitrogen does not have enough time to heat up in the
center of the channel due to the high flow velocity (Fig. 3).
Heating occurs only in the near–wall region of the channel.
The maximum nitrogen velocity in the channel is 1.5 m/s.

Further research is needed, including the use of more
complex channel geometries and experimental studies, to
improve the accuracy and validity of the mathematical
model and optimize the cold spray process. One option for
increasing the complexity of the channel geometry is the
use of spiral channels, which have a more complex shape
than flat channels and allow for modeling of the flow under
more realistic conditions.

4.2. Part 2. Results of numerical simulation in the
ANSYS software system

The flow of nitrogen in a spiral steel tube in a 3D con-
figuration is considered to study the features of gas flow in
such conditions and to optimize the parameters of the tube

to increase the efficiency of gas heating.
Initial parameters: inner diameter of the tube – 0.008

m; length of the tube — 1.154 m; spiral diameter — 0.06 m;
wall thickness of the tube — 0.002 m; initial temperature of
the liquid — 293 K; temperature of the tube walls — 1273 K;
inlet pressure — 2 — 10 MPa; outlet pressure — 0.1 MPa.

The first step of modeling involves constructing the ge-
ometry and mesh. Fig. 4 shows the geometry of the compu-
tational domain with the constructed mesh. The geometry

Figure 4. 3D Geometry of the Computational Domain














































